Author’s response to reviews

Title: Study Design and rationale of 'Influence of Cilostazol-based triple anti-platelet therapy on Ischemic Complication after drug-eluting stent implantation (CILON-T)' study : A Multicenter Randomized Trial Evaluating the Efficacy of Cilostazol on Ischemic Vascular Complications after Drug-eluting Stent Implantation for Coronary Heart Disease

Authors:

Seung-Pyo Lee (splee0624@gmail.com)
Jung-Won Suh (suhjw1@gmail.com)
Kyung-Woo Park (kwparkmd@snu.ac.kr)
Hae-Young Lee (hylee612@snu.ac.kr)
Hyun-Jae Kang (nowkang@snu.ac.kr)
Bon-Kwon Koo (bkkoo@snu.ac.kr)
In-Ho Chae (ihchae@snu.ac.kr)
Dong-Ju Choi (djchoi@snubh.org)
Seung-Woon Rha (swrha617@yahoo.co.kr)
Jang-Whan Bae (drcorazon@hanmail.net)
Myeong-CHAN Cho (mccho@cbnuh.co.kr)
Taek-Geun Kwon (cyclin@naver.com)
Jang-Ho Bae (jhibae@kyuh.co.kr)
Hyo-Soo Kim (hyosoo@snu.ac.kr)

Version: 3 Date: 22 July 2010

Author’s response to reviews:

We have removed the Conclusion subheading, which has been merged to the last part of the Discussion.